Author's response to reviews

Title: The development of a multidisciplinary fall risk evaluation tool for demented nursing home patients in the Netherlands

Authors:

Jacques CL Neyens (j.neyens@irv.nl)
Beatrice PJ Dijcks (b.dijcks@irv.nl)
Jolanda CM van Haastregt (j.vanhaastregt@zw.unimaas.nl)
Luc P de Witte (lp.dewitte@irv.nl)
Wim JA van den Heuvel (w.vandenheuvel@irv.nl)
Harry FJM Crebolder (harry.crebolder@hag.unimaas.nl)
Jos MGA Schols (jschols@vivregroep.nl)

Version: 6  Date: 31 January 2006

Author's response to reviews:

Dear editor,

Hereby I submit the revised version of our manuscript entitled: “The development of a multidisciplinary fall risk evaluation tool for demented nursing home patients in the Netherlands”.

As suggested we made the following revisions:
- table 1: The box format is changed into a table
- table 2: we replaced the vertically written text into horizontally written text and changed the format of this page into landscape.
We hope the manuscript is now ready for publication.

Best wishes,

Jolanda van Haastregt